Author's response to reviews

Title: Health worker preferences for community-based health insurance payment mechanisms: a discrete choice experiment

Authors:

Paul J Robyn (jakerobyn@gmail.com)
Till Bärnighausen (tbaernig@hsph.harvard.edu)
Aurélia Souares (souares@uni-heidelberg.de)
Germain Savadogo (germain_savadogo@hotmail.com)
Brice Bicaba (bicaba_brico@yahoo.fr)
Ali Sié (sieali@yahoo.fr)
Rainer Sauerborn (rainer.sauerborn@urz.uni-heidelberg.de)

Version: 2 Date: 2 January 2012

Author's response to reviews: see over
Prof. Dr. Dr. Rainer Sauerborn  
Chair and Director  
Institute of Public Health  
University of Heidelberg  
INF 365  
69120 Heidelberg, Germany

Prof. Dr. Dr. Till Bärnighausen  
Department of Global Health and Population  
Harvard School of Public Health  
665 Huntington Avenue  
Boston 02115, MA, USA

Africa Centre for Health and Population Studies  
University of KwaZulu-Natal  
PO Box 198  
Mtubatuba 3935, South Africa

Paul Jacob Robyn  
Institute of Public Health  
University of Heidelberg

Nouna Health Research Centre  
Ministry of Health, Burkina Faso  
B.P. 02  
Nouna, Burkina Faso

Phone: (+226)71390734  
Email: jakerobyn@gmail.com

2 January 2012
Dear Editors,

We are herewith submitting our paper “Health worker preferences for community-based health insurance payment mechanisms: a discrete choice experiment” for consideration for publication in *BMC Health Services Research*.

Our study reports the results of the first discrete-choice experiment to elicit health-worker preferences for provider payment mechanisms in the context of a community-based insurance (CBI) in Africa. The study was motivated by qualitative evidence, suggesting widespread dissatisfaction with provider payment in the CBI *Assurance Maladie à Base Communautaire (AMBC)* in Nouna, Burkina Faso. The preferences elicited in the experiment have led to substantive reforms of provider payment mechanisms in the AMBC, including a change in payment frequency, introduction of consultation fees, and a results-based financing component linked to population enrolment and retention in the CBI.

Our study should be of broad interest to health economists and policy makers working on establishing and improving provider payment mechanisms to achieve health-system objectives.

The paper has not been published elsewhere; it has not been accepted for publication; and it has not been submitted elsewhere. We would like to suggest the following colleagues as potential reviewers of our paper:

1. **Dr. Bart Criel**  
   Department of Public Health  
   Institute of Tropical Medicine  
   Nationalestraat 155  
   B-2000 Antwerp  
   Belgium  
   Telephone: (+32)(0)32476 293  
   Email: bcriel@itg.be

2. **Dr. Chris Atim**  
   Health, Nutrition and Population  
   The World Bank  
   Email: catim@worldbank.org

3. **Dr. Robert Basaza**  
   Ministry of Health, Uganda  
   Email: rkbasaza@yahoo.com

Sincerely,

Paul Jacob Robyn, Till Bärnighausen, Aurélie Souares, Germain Savadogo, Brice Bicaba, Ali Sié, Rainer Sauerborn